Author's response to reviews

Title: Colo-pexy as a treatment option for the management of acute transverse colon volvulus: a case report.

Authors:

Mark J Sage (marksage@doctors.org.uk)
Jenan Younis (jenanyounis@aol.com)
Katie E Schwab (katschwab@doctors.org.uk)

Version: 2 Date: 15 November 2011

Author's response to reviews:

Dear Editor

We submit a revised version of the case report taking into account all comments provided by reviewers.

Firstly grammatical changes and vocabulary such as “detorsed” and “volved” have been removed from the manuscript.

The section on aetiology within the discussion has been shortened as recommended. The repetition of radiological findings within the discussion has been omitted. Moreover, a paragraph has been added to discuss the debate between using colopexy versus resection for transverse colon volvulus in cases where the bowel is viable.

Many thanks.